Author's response to reviews

Title: Prevalence of Staphylococcus sp and S. aureus in wounds of hospitalized patients in inland regions of the Northeast of Brazil and associated factors: a cross-sectional study

Authors:

Gilmara Celli Maia Almeida (gilmaracelli@yahoo.com.br)
Marquiony Marques Santos (marquiony@gmail.com)
Nara Grazieli Martins Lima (narinhamartins@hotmail.com)
Thiago André Cidral (thiagoandrecidral@gmail.com)
Maria Celeste Nunes Melo (celmelo@gmail.com)
Kenio Costa Lima (limke@uol.com.br)

Version: 2 Date: 3 March 2014

Author's response to reviews:

To follow the evaluation process, the following adjustments were requested:

1.) Competing Interests
2.) Acknowledgements
3.) Tables must be placed after the References section.

As requested, were added in manuscript the following items: conflict of interest and acknowledgments. Also, the tables were moved to the end of the text (after the references).

We are available for any request or modification on the manuscript.

Kind Regards,
Gilmara Almeida